

This is a repository copy of Integrative objects in sociotechnical contexts: constructing digital well-being with generic epistemology.

White Rose Research Online URL for this paper: https://eprints.whiterose.ac.uk/198455/

Version: Published Version

Proceedings Paper:

Krysztoforska, M., Docherty, N. orcid.org/0000-0001-5945-008X and Biega, A. (2023) Integrative objects in sociotechnical contexts: constructing digital well-being with generic epistemology. In: Schmidt, A. and Väänänen, K., (eds.) CHI '23: Proceedings of the 2023 CHI Conference on Human Factors in Computing Systems. CHI '23: CHI Conference on Human Factors in Computing Systems, 23-28 Apr 2023, Hamburg, Germany. Association for Computing Machinery (ACM), p. 655. ISBN 9781450394215

https://doi.org/10.1145/3544548.3580717

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.





Integrative Objects in Sociotechnical Contexts: Constructing Digital Well-Being with Generic Epistemology

Magdalena Krysztoforska University of Nottingham United Kingdom

Niall Docherty University of Sheffield United Kingdom magdalena.krysztoforska@nottingham.ac.uk n.docherty@sheffield.ac.uk

Asia J. Biega Max Planck Institute for Security and Privacy Germany asia.biega@mpi-sp.org

ABSTRACT

This paper presents a generative approach to interdisciplinary collaboration based on generic epistemology. Informed by the work of philosopher Anne-Françoise Schmid, we introduce the concept of the *integrative object* as a means to reorient interdisciplinary collaboration toward the requirements of the object of research itself, rather than via the requirements of particular disciplinary languages, methods, or operative logics. We show how such an approach is useful for research into sociotechnical phenomena that exceed the boundaries of discrete disciplines and their convergence. We introduce digital well-being as a case study, drawing on the authors' own interdisciplinary collaborative experiences in this area as its empirical matter. From this, and in order to aid future research into similarly complex sociotechnical objects, we then provide practical tools to help those in the HCI community prepare and conduct interdisciplinary research in a similarly generative, non-dogmatic, and non-hierarchical manner.

KEYWORDS

generic epistemology, interdisciplinarity, digital well-being, integrative objects, sociotechnical objects, critical theory, collaborative practice

ACM Reference Format:

Magdalena Krysztoforska, Niall Docherty, and Asia J. Biega. 2023. Integrative Objects in Sociotechnical Contexts: Constructing Digital Well-Being with Generic Epistemology. In Proceedings of the 2023 CHI Conference on Human Factors in Computing Systems (CHI '23), April 23-28, 2023, Hamburg, Germany. ACM, New York, NY, USA, 17 pages. https://doi.org/10.1145/ 3544548.3580717

INTRODUCTION

Contemporary sociotechnical phenomena are often characterized by their multiscalarity, complexity, and heterogeneity, posing numerous challenges for a research landscape delineated by disciplinary boundaries. Consequently, the need to better integrate different forms of knowledge is well acknowledged, especially in fields with interdisciplinarity at their core, such as human-computer interaction (HCI) [9, 86]. However, research objects that cannot be fully



This work is licensed under a Creative Commons Attribution International 4.0 License

CHI '23, April 23-28, 2023, Hamburg, Germany © 2023 Copyright held by the owner/author(s). ACM ISBN 978-1-4503-9421-5/23/04. https://doi.org/10.1145/3544548.3580717

accounted for by a synthesis of existing disciplinary approaches risk being reduced to partial perspectives. This limits the scope for productive engagement with contemporary problems, and threatens the efficacy of interdisciplinary research as a result¹.

This challenge resonates with questions that two of the authors of this paper, a computer science scholar (CS Scholar hereafter) and a critical theory scholar (CT Scholar hereafter), grappled with during a previous interdisciplinary collaboration on digital wellbeing: Why was that so difficult? And, were we actually studying the same thing? Their work together resulted in a paper being accepted at a leading HCI conference, and a shared research agenda that continues to grow. The project could reasonably be considered a success, however during the year-long research stage prior to the publication of their work, the authors encountered several bumps in the road that arose due to fundamental epistemic, ideological, and personal differences extant between them. More importantly, they found that even resolving these differences didn't allow them to orient their understanding of digital well-being towards the future, and did not help them produce new ways of designing 'healthier' computational objects. Essentially, they discovered that digital well-being exceeded the rigid bounds of their disciplinary training and could not be understood through simply combining their different epistemic and methodological frameworks. In other words, they found that a mere synthesis of their disciplinary tools and perspectives was not sufficient to explore digital well-being's complexity as an object of present-day sociotechnical concern.

In scholarly, media, and political discourse, digital well-being is used to describe the ambivalent experiences and effects that various types and levels of technological engagement have on individuals, groups, and societies. Definitions often adopt a normative standpoint, with digital well-being sometimes taken to mean "the impact of digital technologies on what it means to live a life that is good for a human being" [13] (p. 2313), or "a subjective individual experience of optimal balance between the benefits and drawbacks obtained from mobile connectivity" [108] (p. 7). As well as computer science and critical theory, several other fields are invested in the study of digital well-being. These include philosophy [13], media studies [107], clinical psychology [114], media psychology [100], science and technology studies [58], sociology [47], and public health studies [82], to name a few. Each of these fields constructs a different dimension of digital well-being, for example: its philosophical breadth [21], its saliency as a sociological object of inquiry [39], its operationalization as a vector of power [24], its psychological

¹ We use this term to mean any form of working across disciplines, including transdisciplinarity, multidisciplinarity, cross-disciplinarity, etc. We acknowledge that there are ongoing debates related to the specific differences between these terms (see for example Klein [55]), however this is beyond the scope of this paper.

measurability [23], or its technical configuration in computational systems [17]. As we can see from these myriad approaches, digital well-being encompasses multiple aspects of human experience that operate on several scales at once. Consequently, as CS Scholar and CT Scholar learnt first-hand, selecting one particular spatial, temporal, or analytical disciplinary lens through which to study digital well-being loses much that is central to it. For example, how can we present digital well-being solely as a technical concern that better design can fix, when we know that users engage with technologies in unexpected ways beyond their intended use? Do we leave it to clinical psychologists to diagnose the well-being issues surrounding technology, at the expense of understanding technologies' sometimes positive cultural benefits? And can digital well-being reasonably be treated as a large-scale issue for public health and policy researchers, when the effects of technology use are felt on a deeply personal level by individual users themselves? Each of these distinct disciplinary approaches can only provide partial, fragmentary accounts.

When we take a step back from digital well-being, we can notice similar challenges in relation to other contemporary sociotechnical phenomena. For instance, we may think of artificial intelligence, platform infrastructures, information access, sustainable computing, fairness, or data governance. In the field of HCI specifically, interdisciplinarity is highly valued as an approach to issues such as the interface [38], the user [5], or, more recently, the technological enabling of human community, creativity, and action [41]. Indeed, the origin stories and historical developments of HCI are often told as a tale of disciplinary synthesis [9, 86], with some scholars conceptualizing HCI as a design discipline that requires 'radical interdisciplinary dialogue' to problematize, and find solutions to, its areas of concern [112].

This paper resonates with some of the existing debates in HCI. However, we wish to introduce a different perspective to them by drawing on CT and CS Scholar's personal reflections on their collaboration on digital well-being, and on a subsequent encounter with generic epistemology. Generic epistemology is a research and practice paradigm developed primarily by philosopher Anne-Françoise Schmid, and cultivated alongside her various collaborators across different fields (including the sciences, the arts, and design theory) for over two decades. One of the main aims of this approach is to expand and transform relations between disciplines, and to transcend the impasses currently preventing progress in a variety of research areas. Generic epistemology is particularly interested in objects that exceed disciplinary boundaries, and that do not depend on any particular disciplinary framework for their study and interpretation (we can think for example of planetary-scale issues such as climate change, sustainability, or issues pertaining to global governance and trade flows). Schmid refers to such problems as integrative objects, and proposes a non-hierarchical approach to studying them. Specifically, this approach advocates for a logic of interdisciplinarity that doesn't privilege any specific disciplines, theories, or methods. Crucially, it allows for the construction of a heterogeneous research space where disciplinary knowledge can serve to establish the dimensions of objects without reducing them to partial perspectives.

Considering some of the complexities of digital well-being we have presented above, this paper proposes a framing of digital well-being as an integrative object. Using CT and CS Scholar's own experience as empirical texture, we introduce the key pillars of generic epistemology, and argue that a similar approach can help future HCI researchers study other comparably complex sociotechnical problems. The latter parts of the paper offer practical suggestions and prompts that other researchers can experiment with in their own collaborative projects. Section 2 of this paper provides a brief introduction to generic epistemology and its key orientation. Section 3 focuses more specifically on integrative objects, situating the concept in its wider theoretical context, and illustrating its productive potential for studying digital well-being. This section also proposes other objects of HCI research that could be framed in a similar manner. Section 4 discusses generic epistemology's particular approach to the issue of interdisciplinarity in more depth, reflecting on the challenges encountered by CT and CS Scholar in their earlier collaboration as a point of reference. Finally, Section 5 explores the relevance of generic epistemology to questions of design, and presents examples of our own engagement with some of its proposed methods in the context of digital wellbeing. Overall, by focusing on digital well-being as an integrative object, our aim is to introduce generic epistemology to the wider HCI community, and to offer practical suggestions that can help foster productive and non-dogmatic collaborative approaches to other pressing sociotechnical issues.

2 SITUATING GENERIC EPISTEMOLOGY

Generic epistemology is in many ways an unorthodox approach. The origins of the approach are rooted in Anne-Françoise Schmid's work on the relations between philosophy, science, and epistemology [91–93], however generic epistemology is neither a philosophy of science, nor an epistemology in the standard sense. Secondly, although the approach has been cultivated by Schmid and her collaborators across many different fields, spanning the sciences, the arts, and design theory, generic epistemology is not reducible to any of these particular collaborations or contexts. Thus, while generic epistemology is certainly interested in the role that philosophy, science, and design play in the processes of knowledge production and invention, it sees traditionally constructed disciplines as limited in their ability to respond to contemporary problems. It therefore advocates for a remodeling of relations between disciplines, and is chiefly concerned with creating spaces for research and collaboration that do not necessarily follow traditional methods

The key to understanding the orientation of generic epistemology is through its main operator - the generic. Schmid's way of using the term is aligned with that of François Laruelle [59, 60], where the generic serves as an agent of extension and underdetermination. This works to transform any discipline it engages with into raw material that can be used in ways not constrained by the rules and conventions of the source domain. To this end, generic epistemology seeks points of intersection and resonance with other domains, concepts, and problems, and it is not obliged to evaluate its outcomes according to the categories originally prescribed by its source discipline. The generic is invoked to signal that a fragment of a discipline can be implemented in a different context to produce new knowledge, without the wholesale import of the methods

and structures of the original discipline. The generic renders disciplinary concepts autonomous, so as not to replicate the dogmatic and rigid aspects of disciplines that often impede interdisciplinary work. At the same time, the generic recognises the value of disciplinary knowledge and draws on it as its raw material. As such, generic epistemology is neither a specific method, a theory, nor a domain of research. Rather, it should be understood as a general posture and orientation adopted towards research and invention, characterized by its commitment to openness, heterogeneity, and democracy of thought.

The aim of generic epistemology is not to abolish disciplines, but to recognise that many contemporary objects of research require heterogeneous modes of collaboration. Such objects are referred to by Schmid and her co-authors as integrative objects, and they are characterized by not belonging to any specific domain, and by the impossibility of producing them through a synthesis of existing disciplines. It is through an encounter with the concept of integrative objects, and the approach to interdisciplinarity that they entail, that this project began exploring the resonances of generic epistemology with digital well-being, and other sociotechnical phenomena. As outlined in the introduction, the earlier research on digital wellbeing conducted by CT and CS Scholar demonstrated not only the limits of singular disciplinary perspectives in addressing the complexity of this object, but also the limits of straightforward disciplinary convergence. In what follows, we will discuss some of the key aspects of generic epistemology that have proven useful for our understanding of digital well-being, namely: integrative objects, interdisciplinarity without disciplinary continuity, and a generic approach to design. Overall, the aim of generic epistemology, as well as this paper, is not to invalidate existing paradigms of research, but to experiment with an alternative paradigm that shows promise for productive engagement with contemporary sociotechnical objects.

3 INTEGRATIVE OBJECTS

3.1 Integrative objects, complex objects, and boundary objects

Integrative objects are objects of research that pose significant challenges for modes of knowledge production guided by disciplines, as well as for modes of interdisciplinarity predicated on the merging of multiple disciplinary frameworks. One of the key impulses leading to the formulation of this concept was Schmid's collaboration with a biologist, Jean-Marie Legay [67], and their observation that a number of their colleagues from a range of disciplines often struggled to interpret their results. This led them to a hypothesis that perhaps the objects of inquiry and the criteria for interpreting results were not matched. In response to these difficulties, Legay introduced the concept of a 'complex object', characterized by the multiplicity of its components originally studied by distinct disciplines, and requiring the convergence of those disciplines in order to be approached comprehensively.² Drawing on and adapting Legay's work, Schmid

introduces the notion of an integrative object, referring to heterogeneous phenomena that pose challenges for both disciplinary and interdisciplinary research (for example global warming, obesity, or genetically modified organisms).

Schmid differentiates integrative objects from complex objects based on significant methodological insufficiencies - integrative objects cannot be adequately framed or constructed by the same methods that were sufficient for complex objects. Contemporary problems are indeed complex, in the sense that they exceed the limits of any singular discipline, but they are also integrative, meaning that they also exceed the productive capacity of a simple synthesis of multiple disciplines. Schmid writes: "such an object is not the consequence of one discipline along with the collaboration of others. Rather, its dimensions are constituted by fragments of disciplines, and it is a sort of disciplinary 'hole'" [97] (p. 137). In order to productively engage with integrative objects, a simple convergence, overlapping, or synthesis of pre-given disciplines is not enough. The 'disciplinary hole' created around such objects means not only that no singular discipline can speak directly about them, but also that the mode of collaboration between researchers needs to move away from hierarchies and from modes of interdisciplinarity predetermined by disciplinary dogmatism. Instead, researchers need to recognize the fragmentary nature of their disparate areas of knowledge. In this way, the fragments and aspects of expertise that are relevant to the object can enter the problem space as parameters, rather than as authoritative claims. The concept of integrative objects has already been explored in a variety of contexts, including: Robin Mackay's reflections on contemporary art [71, 72], Léo Coutellec's research on Alzheimer's [18], Annie Gentès's work on interactive media technologies embedded in a museum context [31], or Muriel Mambrini-Doudet's research on genetically modified fish [74]. This paper argues that the concept of an integrative object is not only equally useful for the study of digital well-being, but also other objects of HCI research, as we discuss in more detail below.

The concept of integrative objects shares certain characteristics with Susan Leigh Star's notion of 'boundary objects' [10, 68, 105], which has been widely used in a variety of HCI-related contexts.³ However, the two concepts function in different ways and entail different practices. Both of these concepts refer to objects that are heterogeneous, that are constantly evolving, that involve collaboration across disciplinary boundaries, and that motivate continuous development of new research methods. Both concepts are also undoubtedly useful in exploring some of the complexities and challenges of interdisciplinary projects. Nevertheless, there are distinct differences in the overall orientation of the two concepts. Boundary objects are characterized by their interpretative flexibility, and are often intended to facilitate communication between different disciplines. They can be frequently located in the dynamics of collaboration without consensus, where different research communities have their locally-specific interpretations of the objects in question, and collaboration often proceeds by moving back-and-forth between those distinct interpretations. The notion of integrative objects, on

 $^{^2}$ One of the examples proposed by Legay is that of membranes and chemiosmotic theory. Legay sees the separation between biology and chemistry in academia as the reason for the relatively late discovery of chemiosmosis - the knowledge of electrochemical potential of membranes was not within the purview of biologists in the $1960 \circ [66]$.

³ For example: in collaborative design of wearable musial instruments [115], in relation to an AI Playbook for considering potential failures of AI tools before deployment [49], human-AI onboarding materials [15], or to help understand the changing nature of clinical practices influenced by the adoption of computerized systems [116].

the other hand, serves to highlight a different phenomenon. These objects are not produced by disciplines and cannot be contained within them. Therefore, ideas of a shared identity across disciplines, or a disciplinary consensus, are an impossibility. Moreover, integrative objects are non-manipulable, and whilst establishing modes of communication between researchers from different disciplinary backgrounds is certainly an important aspect of generic epistemology, disciplines can only ever offer partial perspectives on these objects. Accordingly, integrative objects cannot be reintegrated back into disciplines and cannot, as such, be fully grasped by them.

While each of the concepts outlined above (integrative objects, complex objects, and boundary objects) can be useful in considering some of the challenges of interdisciplinary research, it is the notion of integrative objects that particularly resonated with our focus on digital well-being, as we will show below.

3.2 Digital well-being as an integrative object

Issues around digital well-being are now highly mediatized, with newspaper reports highlighting the 'addictive' qualities of devices such as the smartphone [102], and whistleblowers decrying technologies' damaging impacts on young people, particularly social media [53]. NGOs, such as the Center for Humane Technology, and charities, such as Mind, lobby for greater awareness of the individual and corporate responsibilities of living well with technology, and national governments, such as the UK's, have proposed legislating against its harmful effects [36]. In response, public media outlets, such as the BBC [87], and private technology companies, such as Google [33], have prioritized digital well-being in their research and development strategies. Overall, these wide and varied usages of the term have led scholars, such as Abeele and Nguyen [109], to describe digital well-being as a "fuzzy concept" (p.176), which resists the constraints of distinct disciplinary approaches, media news cycles, or the proclamations and intentions of interested stakeholders. While the need for the interdisciplinary study of digital well-being has been well articulated [12], it remains an underdeveloped field.

This multiplicity has established digital well-being as a productive concept for several disciplines. However, digital well-being frequently exceeds the epistemic and practical obligations of these disciplines. Clinical psychology, for example, can offer practitioners diagnostic frameworks for pathological usage of certain technologies, such as internet addiction [114], but sometimes struggles to situate problematic technological use, or even its benefits, within its social context. Media psychology, on the other hand, helpfully explores the social situatedness of digital well-being, and the ambivalent value of technological engagement [100], yet elsewhere relies on reductionist behavioral models, such as frameworks of evolutionary cognitive development [107], or neoliberalized social capital [23], to do so. Philosophers of technology provide useful thematic reviews that examine the ethical stakes of digital wellbeing [13], but are not obliged to explore issues to do with technical mediation and the co-construction of healthy tools and their users. Science and technology studies, in contrast, could provide the missing sophisticated sociotechnical analysis of digital well-being [58], but is not always committed to a normative standpoint that can guide what we do with it [63]. In related domains, sociology provides useful topological overviews of the problem space of digital

health in general [47], but likewise is not obliged to critique what is to be found. Finally, public health studies can generate digital well-being insights that are useful for policy makers but, in doing so, these insights could unwantedly result in the perpetuation of operative inequalities depending on the specific socio-political context [82]. While these examples are supplied with very broad brushstrokes, they are included here to illustrate, rather than comprehensively document, the myriad renderings of digital well-being that currently exist in media, governmental, academic, and clinical realms. It is clear that there is no one accepted framework through which to study digital well-being and account for its complexity.

Digital well-being is also an object that escapes the productive capacity of disciplinary convergence. The collaborative work at the intersection of critical theory and computer science that originated this project ran up against precisely this problem. The process of CT Scholar and CS Scholar working together revealed the divergent expectations, methods, and ways of thinking about digital well-being as artifacts of their personal disciplinary training, rather than as optimal or universal responses to the problem space. For example, CT Scholar's training in critical theory and cultural studies had previously led them to treat digital well-being as a historicized discursive formation, paying attention to how it functioned within apparatuses of neoliberal capitalism. Initially for their collaboration with CS Scholar, CT Scholar sought to add technical specificity to their existing research agenda, hoping to examine how the metrification of well-being within computational systems problematically relates to current forms of biopolitics. CS Scholar, on the other hand, approached the problem of digital well-being through the lens of their training in computer science, and in particular in the area of information retrieval, which is chiefly concerned with building information access systems as well as modeling and measuring user behavior and satisfaction. CS Scholar's original intention was to come up with a measurement methodology for quantifying pathological user engagements with a platform. They intended to base the measurement dimensions on the clinical diagnostic framework of digital addiction. For their collaboration with CT scholar, they hoped to use progressive insights from critical theory to curtail the potentially exclusionary ramifications of engaging with such medicalized frameworks. It is clear to see, then, that these disciplinary frames differ in how they conceptualize digital well-being as an object of inquiry. However, it is difficult to say which is the 'correct' approach. Rather, both seem to be apt, yet in a limited way. This realization was the starting point for our exploration of digital well-being as an integrative object, and the catalyst for our further engagement with generic epistemology and its methods (discussed in more detail in the following sections).

3.3 Integrative objects in HCI and adjacent fields

While this paper focuses on digital well-being as a case study, many other concepts in HCI and adjacent fields, including CSCW or FAccT, could be also approached as integrative objects. Looking just at last year's CHI proceedings [2], we can distinguish examples such as fairness, trust, sustainability, child welfare, nudging, and others. Even though a systematic study of these concepts through the lens of generic epistemology is beyond the scope of this paper,

we will now briefly outline its broad relevance using fairness as an example.

Multiple disciplines now study fairness as part of their core research agendas. This includes not only disciplines that have traditionally focused on fairness (such as philosophy, psychology, economics, or law) but more recently also different areas of computer science (including machine learning, or information retrieval), as well as interdisciplinary communities (such as those focused around conferences like CHI, CSCW, or FAccT). These disciplines focus on different aspects of fairness, apply different methodologies, disagree on definitions, and often reject each others' perspectives. For example, computational approaches to fairness focus on mathematical definitions that allow for the algorithmization of fairness [3], critical humanities scholars warn about the harms of these computational abstractions [103], while legal perspectives question the possibility of meaningful automatization of fairness measurement [110].

Moreover, interdisciplinary research spaces often seek to translate and mediate between these different perspectives and demands. Various studies in HCI, for instance, examine user perceptions of fairness that could inform algorithmic designs [37, 106, 111], or probe the needs of various communities of practice when it comes to the prioritization of fairness research directions and driving practical tool development [48, 64]. Among these syntheses of approaches, the community recognizes the need for new ways of working and collaboration. Indeed, Mulligan et al. have experienced many of the challenges we touch on in this paper and, in response, developed a discursive framework for initiating interdisciplinary collaborations on fairness [77]. Whilst the increasing interest in approaching concepts such as fairness in more collaborative and innovative ways attests to the growing recognition of the limits of disciplines, many of these approaches ultimately rely on disciplinary conventions of rigor in order to evaluate their results. As we discuss in the next section, generic epistemology proposes that reverting to disciplines in this way is not compatible with integrative objects, and a different mode of interdisciplinarity is needed.

4 GENERIC EPISTEMOLOGY AND INTERDISCIPLINARITY

The way in which generic epistemology approaches the question of interdisciplinarity is deeply connected to the concept of integrative objects. Since these objects are not given within disciplines, and cannot be constructed through a convergence of disciplinary approaches, they require a reconfiguration of both the role of disciplines and forms of interdisciplinarity. This realization developed over time, as Schmid and her collaborators kept noticing an increasing number of scientific debates lacking satisfactory solutions for given problems, and relying on deficient conceptual frameworks. Even increasingly common interdisciplinary practices were often stuck at a dead end. The transfer of epistemological frameworks from one discipline to another often proved inadequate, and the differences between the languages of separate disciplines created additional difficulties.

These observations on the challenges of constantly evolving research landscapes are of course not unique to generic epistemology. Numerous scholars have pointed out the increasing importance of interdisciplinarity, with some theorizing it as a shift from 'Mode

1' to 'Mode 2' of knowledge production, moving from a homogeneous view of science to a diversification of research application and evaluation [32, 78]. Others have also articulated the need for a critical interdisciplinarity that does not replicate disciplinary modes of research, but instead expands in line with broader dynamics of knowledge production [29]. Disciplines with interdisciplinary research at their core (such as: science studies [7], ecological economics, technology assessment, science and technology studies [54], or indeed HCI [9, 86]) can be also seen as part of this expansion. The status of HCI as a discipline is itself contested [16, 22], with different tensions at play. Scholars have discussed the different implications of conceptualizing HCI as craft, applied science, and engineering [70], as well as the tension between HCI's relationships with both science and design [85]. Alongside the anxieties around the status of HCI as a discipline, some scholars see these frictions as a positive dynamic, motivating innovation within the field [8, 9].

Nevertheless, generic epistemology identifies modes of interdisciplinarity that rely on a synthesis of, or a friction between, disciplines as insufficient. Interdisciplinary collaborations often import rigid disciplinary frameworks and logics into their research space, leading to potential deadlocks. This assessment resonates with a broader discourse on the different logics of interdisciplinarity [4, 52, 55], and on the challenges related to interdisciplinary work (such as institutional barriers [52], barriers to funding [81], cognitive obstacles [73], and divergent worldviews constituted by the different metaphysical and epistemological commitments of different disciplines [80]). In HCI specifically, the community has made continuous efforts to develop practices that can help overcome obstacles in interdisciplinary collaborations. As will be discussed in more depth in Section 5, 'action design research' [101], 'contactzones' [19], and practices of 'co-design' [89], for example, have been developed to facilitate the study of 'wicked problems' [88] that defy disciplinary conventions. Elsewhere, some HCI scholars propose 'disciplined transdisciplinarity' as a model for interaction design [76], whilst others advocate for a practice of ongoing critical self-reflection [25]. The problem of interdisciplinarity has also been presented as a problem of clear communication, with some arguing for a 'Liberal Arts of HCI' education that supplies practitioners with skills to foster better disciplinary relations in the future [112]. Other scholars propose spaces for equitable conversations about the challenges of interdisciplinary research [50], as well as protocols for developing trust between researchers [113].

This multiplicity of approaches to collaborative research is certainly seen by generic epistemology as positive, since it allows for increasing openness of research, and for more dialogue between different areas of knowledge. Nevertheless, approaches that see contemporary objects⁴ as givens, and as knowable by disciplines (or their convergence), are ultimately different from the orientation proposed by Schmid and her collaborators. A productive engagement with integrative objects requires a generic space where different fragments of knowledge and 'non-knowledge' can create new relations, without a wholesale import of pre-existing disciplinary logics, and without recourse to disciplines to evaluate research outcomes. This key point differentiates the

⁴ The term 'contemporary object' is used throughout this paper as equivalent to 'integrative object', in line with Schmid's use of these terms; see for example Schmid [94].

approach of generic epistemology from numerous other interpretations of exchanges across incompatible research paradigms. For instance, one commonly used metaphor is Peter Galison's concept of a 'trading zone', which is used to describe research communities working together that often don't share the same scientific language or even the same interpretation of the meaning of the collaboration [30]. The concept has been used in a wide variety of interdisciplinary contexts, not least in relation to socially-relevant technologies [34], sociotechnical networks [35], or to describe the HCI community more broadly [9]. While the approach of Schmid and her collaborators certainly resonates here, generic epistemology is not primarily concerned with disciplinary languages and negotiations between different local interpretations of an object of research. Instead, it advocates for the creation of a space outside of these dynamics in order to study integrative objects. Moreover, while generic epistemology does not overlook the problem of incommensurability in research, its approach to it is also different from that of Feverabend [27] or Kuhn [56]. This is because generic epistemology is not invested in comparisons between theories, nor does it participate in a process where a new paradigm makes the previous one obsolete. Instead, the generic enacts what Schmid calls 'a paradigm shift without crisis', which occurs without the need to disavow disciplines. The generic works here to create an 'outside discipline' to be filled with new modes of inquiry.

The non-hierarchical nature of generic epistemology allows for all paradigms to continue, and enables a process where fragments of existing knowledge can be used as raw material for new modes of thinking, unbounded by the rules of their source paradigm. This approach is motivated by what Schmid calls 'Poincaré's criterion', inspired by the logic of invention she sees as underpinning the work of Henri Poincaré⁵, and his interpretation of scientific generalization. Generic epistemology posits that in the absence of disciplinary criteria for scientificity, the rigor of an interdisciplinary practice cannot be assessed in relation to disciplinary coherence. In its absence, Schmid proposes the logic of invention as an alternative means to establish the rigor of a collaboration. As written by Schmid and Mambrini-Doudet: "Poincaré's criterion combines decomposition and hypercompatibility. It supposes a space that is common to a series of disciplines, which is ordered in a way that makes possible the decomposition of disciplinary propositions. In this space, if we construct a model, we can construct an infinite number of others, none of them having more descriptive force than the other. The models make the knowledge hyper-compatible but they are not the only possible orders" [98] (p. 122; our translation). Since disciplines can only constitute specific dimensions of integrative objects, and offer only partial perspectives on them, such objects require a more open approach to collaborative research. Schmid and her collaborators propose the term collective intimacy⁶ to describe the conditions for collective research that encourage interactions between knowledge and non-knowledge. This works to form "a collective cogito" [99] (p. 38). Unlike boundary objects,

therefore, integrative objects are not passed on from one discipline to another (or shared between them). As a result, integrative objects require different, *generic* conditions for collaborative research and invention. The following section will now present some of these methods in more detail.

4.1 Digital well-being and making visible the components of disciplines

Generic epistemology doesn't aspire to be a rigid framework and certainly doesn't take a dogmatic stance on suitable methods and approaches. Nevertheless, several general practical recommendations for fostering interdisciplinary research have emerged through Schmid's various collaborations. Principally, generic epistemology posits that a collaborative practice that doesn't impose a hierarchy of knowledge, and isn't constrained by disciplinary conventions, requires space for new relationships between disciplines to emerge. To enable this, Schmid encourages experimentation and creating space for 'theories of intermediate scope' [98] (p. 37; our translation) that are particular to the object of concern. A key step towards this process is for researchers to recognize the limits of their disciplines and to consciously map their internal landscapes by breaking them down into their separate components. Schmid and Mambrini-Doudet outline six key components of the internal landscape of any discipline that can help this orientation. These are: intellectual apparatus, temporality, modes of organization, a regime of scientific recognition based on appropriation and critique, a social system for producing peers, and an implicit knowledge of science [98] (pp. 23-24; our translation). The internal balance and relative differences between these elements are what differentiates and individualizes disciplines. In what follows, we discuss the different components outlined by Schmid and Mambrini-Doudet in turn, illustrating them with examples from CT and CS Scholar's collaboration on digital well-being. We hope to show how engaging with this way of understanding disciplines sheds new light on the subjective experience of personal interdisciplinary collaborations.

• Intellectual apparatus – the main goals and intellectual motivations of a discipline.

Through their collaboration, CT and CS Scholar discovered that their respective disciplines had vastly different ways of conceptualizing digital well-being, and vastly different motivations for doing so. As already mentioned, CT Scholar initially sought to examine digital well-being as a contemporary location of biopolitics. This can be seen as a reflection of critical theory's overarching motivation to clarify oppressive power relations, and forms of subjugation, operative in the world at differently technologized junctures. However, through the encounter with CS Scholar, CT Scholar began to see the limits of this apparatus of thought. CT Scholar realized that although their training allowed them to clarify and critique the broad capitalist functioning of digital well-being, practical suggestions to improve the situation were not always a priority, or even a possibility, within the field. Conversely, CS Scholar began the collaboration with an intention to

⁵Schmid's own style of writing and thinking has been significantly influenced by Poincaré, whose work she devoted her first monograph to [92].

⁶The term was first coined by ethno-psychiatrists Lucien Hounkpatin, Henny Wexler-Czitrom, Avner Perez and Laurianne Courbin as a non-dogmatic therapy paradigm that doesn't simply apply psychiatric knowledge to patients, but instead fosters multiple relations between this knowledge and patients' own socioculutral experiences [51].

create a new metric for digital well-being, in order to help rationalize and standardize its future study and technical application. However, through their encounter with CT Scholar, CS Scholar started to clearly see how this drive was not universal to all disciplines. Rather, it was a reflection of their trained computational instincts to design concrete solutions to digital well-being, which often entailed uncritical application of imported models and concepts.

 Temporality – the timescales of research within a discipline, as well as the evolution of a discipline and of its ideas over time.

One of the possible interpretations of this category refers to practical matters associated with working schedules, publication timelines, and the expected duration of projects in each respective field. The differences between critical theory and computer science in this regard was most obviously felt when it came to the timelines associated with publication. For instance, CT Scholar had to familiarize themselves with the conventions of peer-reviewed conference proceedings and the fairly strict deadlines for submission to large computer science venues at set times of the year. Working toward a deadline for submission was not something CT Scholar was used to, as submitting to the humanities venues where critical theory scholars usually publish occurs on a rolling basis. Moreover, the revise and resubmit process of CS venues similarly followed a standardized timeline, which was again novel to CT Scholar, considering the months, and sometimes years, that a humanities paper could languish in peer-review and editorial processes, which change from journal to journal.

• Modes of organization – the main operative logics and practices within a discipline.

During weekly project discussions, certain practical issues had to be resolved. Firstly, CT and CS Scholar had to decide what technological tools to use to share ideas and draft their work. This was not as straightforward as it may seem. For example, CT Scholar preferred MS Word and its privileging of prose, while CS Scholar used LaTex and Overleaf, valuing their greater capacity for tabulation and automatic reference numbering needed in ACM formats. Another key issue was the publication venue. CT and CS Scholar had to discuss where their analysis and findings would be best placed, and consider what disciplinary conversations would 'hear' their interventions. As we shall show below, this had significant implications for the job security for both.

Regime of scientific recognition based on appropriation and critique – the way a discipline constructs itself in terms of what it embraces and what it rejects.

A key point in CT and CS Scholar's discussions was how best to align their ideological expectations of the project. This chiefly meant resolving the tension between the desire to design metrics of digital well-being versus analyzing such metrics as vectors of normalizing contemporary power structures. Critical theory's distaste of easy fixes ran up against computer science's deeply rooted need for projects that have practical, measurable outcomes - a tendency often termed 'technological solutionism' [104]. The tension was particularly visible in the writing of the 'Design Implications' section of the paper, which was a key priority of the computer science venue they had chosen to target. Here, the level of concreteness of the design implications was subject to particularly intense negotiations. It was at this stage, when it came to actionable 'solutions' to digital well-being, that the differences between CT and CS Scholar were most clearly felt.

 A system for producing peers – the social dynamics of the research environment within a discipline.

Interdisciplinary research is highly valued by the European higher education institutions and governmental funding bodies that both CT and CS Scholar wished to join and engage with after their postdoctoral positions.⁷ For CT Scholar in particular, coming from a United Kingdom arts and humanities background, being able to demonstrate collaboration with fields that were perceived to be more instrumentally valuable, such as computer science, granted a sense of legitimacy to a threatened research area [6]. In this way, interdisciplinary research projects offer early career humanities researchers a valuable foothold in the increasingly competitive academic world. For CS Scholar, however, a different set of institutional logics was at play. For example, while computer science as a field shows increasing interest in hiring scholars with interdisciplinary interests, it remains to be seen whether faculty evaluation and promotion criteria have been sufficiently adapted to account for interdisciplinary outputs. There are issues around how to value and compare different publication venues, or how to account for the slower time frames of working and publishing across disciplines. For computer scientists, it could be considered less risky to prioritize publishing in CS venues, and only engage with other fields as a source of inspiration and a guiding input.

• Implicit knowledge of science – the way in which a discipline establishes its criteria for rigor.

As discussed earlier, CS and CT Scholar became aware of each other's disciplinary differences early on in their collaboration, when the conflicting nature of their initial goals for the project became apparent. The process of working together enabled the researchers to venture outside of their respective disciplinary boundaries, however the fact that even interdisciplinary research is often evaluated by disciplinary standards created a significant problem. Whilst both CT and CS Scholar started seeing

⁷ For example, two of the largest public funders in this region, the European Research Council and the UK Research and Innovation non-departmental government body, both emphasize the value of interdisciplinary research.

their individual perspectives on digital well-being as only partial, they still felt obliged to produce research outcomes that conform to their respective discipline's standards of rigor. As a result, the tension between metrification and critique could not be fully resolved.

The above categories are able to help map the internal landscapes of disciplines. According to generic epistemology, doing so helps distinguish the key modes of knowledge production that disciplines engage in. This, in turn, opens up the possibility of their underdetermination and of compatibility between different fragments of knowledge. Schmid and Mambrini-Doudet distinguish six main modes of knowledge production: dissociating, appropriating, linguistic, cultivating, illusory, and variety [98] (pp. 24-27; our translation).⁸ Exploring the dynamics of research in this way can be useful in recognizing their insufficiency. It can help us recognize our own orientation as partial, and conditioned by the specific conventions of our own disciplines. In what follows, we will discuss each of these modes in more depth, drawing on the experiences of CT and CS Scholar to illustrate their workings.

• **Dissociating mode** – refers to the way the discipline self-evaluates its own progress based on logics of justification and invention. Recognizing the particular expression of this mode within a discipline allows us to reposition it, and place it within a plurality of other approaches without undermining its efficiency.

In relation to this mode, we can see how the original motivations for CS and CT's interdisciplinary collaboration could not have been further apart: one researcher wanted to critique the measurement of digital well-being, the other wanted to construct new metrics for its administration. Upon this recognition, the project progressed beyond the limitations of each respective disciplinary approach, ultimately combining in a way that could not be understood through each approach singularly conceived.

• **Appropriating mode** – relates to the ways in which a discipline distinguishes itself from others based on what it appropriates and what it critiques.

Critical theory adopts overarching theoretical frameworks through which to examine the world from the perspective of the oppressed and subjugated. Frequently, capitalism, colonialism, or power are invoked as such overarching concepts that link disparate empirical analyses together. Conversely, critical theory rejects concepts that are used to maintain unequal social systems that privilege certain groups over others. For example, this could include conceptual tools associated with neoliberal differentiation and

administration, such as social capital. In the realm of computing and data science, statistics and machine learning seek to distinguish each other in ongoing debates. Some argue that the difference lies in typical tasks and methods the fields engage with [14], while some claim that the two are no different except perhaps for marketing [1].

• Linguistic mode – relates to the various layers of the specific language used by a discipline, including specialized vocabulary and specificity of knowledge (theorized by Galison as scientific 'pidgins' and 'creoles' [30]). It clarifies the stakes involved in developing modes of compatibility with other disciplinary dialects (encounters between vastly different disciplinary contexts can lead to 'culture shocks').

When CT and CS Scholar started discussing their intentions for the project, they quickly realized that they were often speaking at cross purposes, using the same terms to refer to vastly different objects or concepts, or using jargon that had no disciplinary equivalent for the other. The meaning of the terms and concepts that grounded CT and CS Scholar's respective research often dissolved in their attempted translation, along with their usefulness. To use a slightly unorthodox comparison, we may think of these initial meetings in terms of a 'first contact' scenario, whereby distinct alien species attempt to communicate with each other across their radical communicational differences. In trying to stabilize terms, both scholars revealed implicit disciplinary values that had, until then, slipped below their conscious consideration.

• Cultivating mode – makes visible the 'thickness' of a discipline, characterized by the density and rigidity of the logics of cultivating new ideas within the discipline. This determines to what extent a discipline might want to either close itself off or enable a space of exchange with other forms of knowledge.

The experiences of CT and CS Scholar attest to the generally porous nature of both critical theory and computer science, despite their sometimes dogmatic tendencies. Aside from some of the difficulties outlined above, this year-long project led to a successful completion and motivated the next stage of the collaborative exchange. By exploring computer science and critical theory through the lens of the notion of cultivating mode, both researchers were able to

⁸ Schmid and Mambrini-Doudet use the term 'machine' instead of 'mode' here, drawing on the work of Jacques Lafitte, and Gilles Deleuze and Félix Guattari. Given that the term 'machine' might have different connotations for the audience of this paper, and that the work of Lafitte and Deleuze and Guattari is beyond the scope of this current discussion, we chose the term 'mode' in order to preserve the same dynamic, production-connoting meaning, without adding an unnecessary layer of opacity to the concept.

⁹This framing of interdisciplinary communication is of course not too dissimilar from the anthropological idea of first contact and its colonial history (also directly relevant to Galison's theorization of trading zones and the communicational issues they entail). Nevertheless, exploring the linguistic mode of disciplines more speculatively, in terms of xenolinguistics rather than anthropological linguistics, enabled us to interrogate the dynamics of interdisciplinary exchange on an even more fundamental level. How would we approach the task of collaborative research if we could not make any assumptions about the key building blocks of a discipline we were interfacing with? And, more importantly, how would we go about explaining our own disciplinary culture in such a scenario? Sending messages into outer space with the hope of establishing contact with an alien species has a long and rich history, and points to numerous challenges of communicating over vast distances, and without the possibility of making any assumptions about alien interlocutors. Many of the messages sent into space have relied on visual and mathematical ways of encoding information, considering these more universal than natural language. For a detailed account of this history, see [79].

recognize their respective disciplines as responses to particular societal and technological shifts, and as mediators between disciplines that predate them. This put in perspective the particular ways in which these disciplines have been historically constructed, and, consequently, helped shed light on how the knowledge they produce can be entered into 'spaces of interdiscipline'.

 Illusory mode – also related to this potential for exchange or isolation, refers to a disciplinary culture as marked by the illusion of wholeness. This is manifested in constructing macroscopic theories or limiting the interactions with other disciplines to superficial exchanges such as criticism or projection of one onto another.

While an illusion of wholeness is perhaps easier to observe in the case of philosophy, which often grants to itself the authority to speak on behalf of other domains ¹⁰, CT Scholar noticed similar tendencies in their own discipline during the collaboration. Critical theory tends to perceive most contexts concerning society as material ripe for critique, and being aware of this predisposition allowed CT Scholar to recognize some of their initial instincts in approaching digital well-being as motivated by this particular illusion of wholeness, rather than as necessity.

• Variety mode – helps to understand a discipline as a heterogeneous system. The constant evolution of disciplines points to the implicit understanding that disciplinary knowledge is insufficient, and that there are areas of non-knowledge; disciplines do not therefore fully cover the scientific horizon.

Reflecting on their initial motivations for starting a collaborative project enabled CT and CS Scholar to notice some of the differences in how their respective disciplines value heterogeneity of research and knowledge. Critical theory, similarly to many other disciplines within the humanities, increasingly recognizes its own local character and scope, which is also why interdisciplinary work is highly valued in the humanities. In computer science, institutional logics have not yet adapted to evaluating interdisciplinary work on an equal footing, however researchers show increasing interest in diversifying the variety of knowledge they engage in. An analysis of these broader dynamics motivated CS and CT Scholar to question even further the processes within their disciplines that distinguish areas of knowledge and non-knowledge.

4.2 Internal landscapes of disciplines and modes of knowledge production – prompts for an interdisciplinary dialog

Based on our engagement with generic epistemology, as well as the retrospective reflection on the experiences of CT and CS Scholar, we propose that a dialog focused on examining the limits of disciplines involved in a collaboration can open up the possibility of a non-hierarchical research space. We suggest that this type of reflexivity can help curtail the transference of disciplinary dogmatism into a

collaborative project. To help guide researchers in this process, we have developed a semi-structured script based on the components of disciplines outlined by Schmid and Mambrini-Doudet. These are reflected in Table 1 and Table 2. We hope that by engaging with these prompts, researchers can help themselves shed light on the incompatibilities in values and incentives early on in a collaboration, and, more importantly, help steer the project towards a mode of interdisciplinarity without disciplinary continuity.

Table 1 presents an overview of the elements of disciplinary landscapes, our reading of why understanding these elements is crucial in the context of an interdisciplinary collaboration, and our proposed inventory of example prompts that researchers could include in their initial exploratory dialog. Table 2 proposes a series of prompts intended to help identify the particular forms of expression of the modes of knowledge production within any discipline. The scripts do not seek to be dogmatic or exhaustive; our goal is to provide intuitions and example interpretations of the framework, leaving their application and adaptation up to individual researchers and the specifics of their projects. Importantly, we should note that the process of engaging with these scripts does not itself constitute a method for working with integrative objects. Rather, engaging with the prompts is just one of the initial steps a researcher could take to avoid importing disciplinary dogma into an interdisciplinary research space. Methods that are more directly intended for working with integrative objects will be discussed in more detail in the next section.

5 GENERIC EPISTEMOLOGY AND DESIGN

Generic epistemology is also interested in questions of design, especially as an exploration of interactions between philosophy and processes of invention. An integrative object is posited in the context of design as an unknown object, or 'object X' [95], the properties of which are often redistributed in unexpected ways. Much of this aspect of generic epistemology has been developed by Schmid in collaboration with Armand Hatchuel, a design theorist, and some of the methods they engage with are related to Hatchuel's C-K Design Theory (which he developed together with Benoît Weil) [42-44]. This theory, also called concept-knowledge theory, posits design as a process of extension, one involving constant movement between the space of concepts (C) and the space of knowledge (K). Similarly to how generic epistemology emerged as a response to the observed limitations of conventional forms of disciplinary research, C-K theory developed in response to certain limitations of the dominant approaches to design observed by Hatchuel. In his view, the scope for invention in design is often limited by an excessive focus on problem solving, as well as by classic design theories being too reliant on specific domains of implementation and too separate from theories of creativity. Generic epistemology is interested in design primarily as a non-dogmatic process of invention and extension, allowing for unexpected conceptual formulations. The key driver of this process is again the notion of the generic, and Schmid and Hatchuel developed a number of general recommendations for creating generic-like conditions for this process to flourish. They outline five main procedures [97] (pp. 136-137), which rely on extracting concepts from their original disciplinary contexts in order

¹⁰ As repeatedly pointed out by Laruelle, see for example: [61, 62].

Table 1: An inventory of example prompts for a first contact interdisciplinary discussion based on the elements of landscapes of disciplines in generic epistemology.

Elements of the landscapes of disciplines according to generic epistemology	Reasons for understanding the element in an interdisciplinary collaboration	Example discussion prompts
Intellectual apparatus	Understand the main intellectual motivations of the discipline.	What are the research goals of your discipline? What is your current research agenda? What are some research questions your discipline is interested in directing at the object?
Temporality	Understand the general evolutionary trajectory of the discipline and its ideas over time. Understand to what extent the discipline values innovation over preservation of knowledge.	What is the typical publication timeline in your field? How quickly does published research lose relevance in your field? In a typical paper in your field, how much weight is there on aspects of novelty vs building on existing scholarship? What is your personal timeline for this project? By when would you like to (need to) have this project wrapped up? Is the study of this object in your field considered traditional or innovative?
Modes of organization	Understand the main operative logics, methods, and practices used within the discipline.	What are the typical publication venues in your field? How would you attempt to study the object? What methods would you apply? How do you personally structure a project? How do you typically organize your work?
Regime of scientific recognition based on appropriation and critique	Understand how the discipline constructs itself in terms of what it embraces and what it rejects. Understand which perspectives and pursuits are valued within the discipline, and which are critiqued.	What kinds of objects and problems does your discipline consider worthy of research? How do you judge the quality of a paper in your discipline? What kind of examination of the object would your field find outrageous?
A system for producing peers	Understand the career progression requirements of collaborators, and the broader social dynamics of the research environment. How are people evaluated for hir in your field? In your field, how of ferent seniority typically contrib. How could this project help you career right now? What brought cipline? Who are the people in have studied the object?	
Implicit knowledge of science	Understand the state-of-the-art knowledge in the field. Understand what constitutes rigor in the discipline.	What are the research questions at the frontier of knowledge in your field? What are some contemporary 'must knows' in your field? What makes a paper in your discipline good? What is the dominant research paradigm in your field? What methods and approaches are considered rigorous in your field?

to reconfigure them and produce them anew. These include: allowing for the autonomy of concepts; approaching disciplines as parameters rather than content; writing experimental texts in order to construct hypotheses on objects and their properties without the need to restrict the objects to their properties known in the present; constructing cartographies of research where all the components are in dynamic relationships with disciplines; and forging connections between scientific and extra-scientific domains.

The pursuit of experimental methods of invention also led Schmid and her collaborators to develop the technique of *fiction* and 'the method of without' [95], which can help reimagine the problems at hand, and hypothetically design alternative ways of approaching

them. The concept of fiction is used to signal generic epistemology's orientation towards the future and its embracing of the unknown. Examining objects from the point of view of their futural extension is a challenge that traditional disciplinary methods often struggle to grapple with. Accordingly, Schmid operationalizes the notion of fiction to emphasize that integrative objects cannot be reduced to what is currently known and accepted by disciplines. The technique of fiction approaches objects as not fully realized, and searches for their invariant characteristics across different fragments of knowledge.

Table 2: An inventory of example prompts for a first contact interdisciplinary discussion illuminating different modes of knowledge production in a discipline according to generic epistemology.

Mode of knowledge production according to generic epistemology	Reasons for understanding the mode in an interdisciplinary collaboration	Example discussion prompts
Dissociating mode	Understand how the discipline thinks about progress and innovation. What will the collaborators want to achieve? What counts as a novel contribution?	What would have to be included in a paper for it to count as a novel contribution in your field? What would be some natural next steps in the study of the object in your field?
Appropriating mode	Understand which methods and approaches col- laborators would consider valid and appropriate in the study of the object. Which methods and approaches collaborators would be resistant to using.	Which research methods or discourses does your field find (un)acceptable in the study of the object? Which of the research methods/discourses in my field does your field find (un)acceptable?
Linguistic mode	Learn the terminology a field uses in the study of the object and related topics. Are there any ideological stances encoded in the language and jargon of the discipline? Can it lead to 'cultural shocks'?	What is the broader set of terminology that your field uses when discussing the object and related topics? When I use the following terminology related to the object, what do you hear, and how would you understand it?
Cultivating mode	Understand the ways in which the discipline forms its operational logics and rigor. Understand the conditions which allow for knowledge exchange with other disciplines.	How do you typically share your research with others in your field? What are some of the events where scholars typically meet? What are the main publication venues? How does peer review work in your field? Is it common for scholars in your field to collaborate outside of the field? What do these collaborations typically look like (topics, processes, outcomes)?
Illusory mode	Understand how the collaborator's discipline typically perceives its own limits (if at all) and its compatibility with other disciplines. Does the discipline consider itself universal in scope?	How broad is the scope of your discipline? How does your discipline relate to mine and other disciplines? Is there something lacking in your disciplinary framing of the object?
Variety mode	Understand what the discipline sees as its own constitutive elements and how they evolve. Understand what the discipline acknowledges not to know or what is not included within its purview. Understand what productive potential the discipline sees in an interdisciplinary collaboration.	In what ways has your discipline changed in the last few years/decades? What are the most dynamic aspects of your discipline? What does your discipline explicitly not study? What does it acknowledge not to know? How can your discipline benefit from interacting with my discipline?

Generic epistemology's investment in methods of invention could be of particular interest and relevance to the HCI community, given the field's affinity with design. Scholars such as Daniel Fallman [26] have described HCI as a 'design-oriented field' (p.225), and Wright et al. [112] frame HCI as aligned with designerly disciplines and their creative problem solving. Here, HCI practitioners are encouraged to bring their designerly attributes to the study of contemporary sociotechnical objects, and problematize them in a way that compliments, and may help re-imagine, other disciplinary expertise working on the project. This ability of design thinking, conceptualized by Buchanan in terms of the 'doctrine of placements' [11] (p.8), is seen as able to offer new perspectives, which can lead to new insights, and solutions, to emerge as a result. This approach to problem-solving, (dis)placement, and contextual relevance has led some design theorists to frame designerly thinking as a 'meta-disciplinary methodology' [69] (p.35). HCI literature also often invokes the value of 'designerly ways of knowing' as developed by Nigel Cross [20], offering a transcendent vision of

design as a research culture fundamentally tasked with creative and practical solutions to problems across almost unlimited domains. Furthermore, the notion of the 'wicked problem' has proven useful in HCI to conceptualize complex sociotechnical issues that require innovative and multi-disciplinary design interventions [57, 65, 101]. The term is often imported from design theory, and defined as: "problems that are ill-formulated, potentially unbounded, open to alternative formulations with no obvious means of choice or stopping rule and are particular to a situation, setting or context [11] (p.9). In an influential paper, Zimmerman et al. [117] argue that these types of 'under-constrained problems' are natural concerns of HCI due its span across a wide swathe of interrelated technical, human, and social processes. The authors advocate a move away from strictly instrumentalized approaches to their solution, and recommend a more holistic mode of intervention.

Whilst generic epistemology is not in itself a design theory, it certainly resonates with numerous design-related discussions in HCI literature. The welcoming of new ways of thinking, and the general

position of openness that characterizes generic epistemology can be also attributed to many of the perspectives outlined above. Nevertheless, there are also some distinct differences that are important to note, and while designerly ways of knowing are often, if not always, concerned with finding solutions, the methods of working with integrative objects are not predicated on the same need. Simply put, there is no obligation to find timely practicable solutions to problems posed by integrative objects, even though actionable responses to research findings are very welcome outcomes of such collaborations. It is important to state that this outcome cannot be known or even planned from the outset, and any collaboration could result in the identification of more problems 'all the way down' [90]. A generic approach to design is primarily concerned with decomposing the existing modes of knowledge production and invention, in order to enable their new modes of compatibility.

Generic epistemology's lack of obligation to solutionism or disciplinary maintenance perhaps has stronger resonances with more recent work in HCI that foregrounds non-instrumental, non-hierarchical, and progressive approaches to sociotechnical design [83]. Anna Croon [19], for instance, puts forth an idea of 'thinking with care' in HCI that brings together diverse voices in the design process according to feminist principles of situated knowledge, incommensurability, and critical sensibilities. Drawing on feminist STS theorist Donna Haraway's concept of the 'contactzone' [40], which itself was originally coined by literary theorist Mary Louise Pratt [84], Croon argues that the design process of sociotechnical objects must incorporate a wide variety of stakeholders if they are to curtail the entrenchment of existing inequalities and systematic oppression through their operationalization. Such an approach is radically equitable, and differences between stakeholders may ultimately be irreducible. A set of related practices emerging from HCI are encapsulated in 'co-design spaces', as understood by Sanders and Westerlund [89]. Here, the authors seek to create the conditions through which differently embodied knowledge can come together to imagine new design futures for HCI in a shared, physical space. Whilst the general orientation of these approaches is in some ways parallel to that of generic design, the latter emerged from a different context, and uses different conceptual material. As a result, it proposes a range of methods and practical tools that might not yet be familiar to the broader HCI community. In what follows, we outline some of these methods, and illustrate them with our own examples related to digital well-being.

5.1 Digital well-being in generic space

A number of the aforementioned procedures for producing the generic served as the basis for developing our own approach to digital well-being, and guided us in constructing a generic space for this particular research object. More specifically, our project explored the possibility of generating new research hypotheses and perspectives on digital well-being through the process of disciplinary underdetermination. In Section 5.1.1 we discuss this method in more detail, and provide an example of how critical theory and computer science can be decomposed into their specific parameters. We then use these insights to construct hypotheses and partial models of digital well-being that can be developed further in the future. Moreover, the method of fiction proposed by Schmid and

her collaborators has also proven fruitful for our understanding of digital well-being. Section 5.1.2 explores possible applications of this method, and proposes ways of approaching digital well-being as a hypothetical object of design.

Our way of working with these methods was rooted in Schmid's and Hatchuel's recommendation of writing *experimental texts*. Experimental texts help create models, sketches, and logical maps, diagramming the parameters of disciplines relevant to the objects of inquiry, and helping to identify the points of intersection between disciplines, as well as their blind spots. They are experimental in the sense that they often do not follow traditional formats of academic or scientific inquiry, and they do not require a definitive research outcome. These are texts "from which a certain number of hypotheses can be constructed for the work that follows" [96] (p. 110; our translation). What we present below can also be understood in terms of experimental texts – our examples are not meant as definitive or prescriptive. Their primary function is that of *hypotheses*, which we intend to build upon in our further research.

5.1.1 Decomposing disciplinary perspectives on digital well-being. When singular disciplines or their composites claim authority over a specific object of research, the dimensions of the object are restricted to the space delineated by operative disciplinary parameters. Conversely, generic epistemology proposes a process where disciplines are decomposed into their key components in order to extract them from their disciplinary context, and place them within the generic space of the object. As a result, the features and dimensions of the object previously considered exhaustive by the disciplines are revealed as partial, and the object is rendered unknown or underdetermined. This enables new interactions between those partial perspectives, unbounded by the rules of the disciplines that produced them. There instead emerges a multiplication of "disciplinary dimensions, while [also] allowing, where they do not meet or intersect, constitutive zones of unknowing" [98] (p. 68; our translation).

In our own project, we experimented with this process by using critical theory and computer science as means to sketch out possible dimensions of digital well-being. The first step of this process was facilitated by the tools presented in Tables 1 and 2. The exploration of the internal landscapes of critical theory and computer science allowed us to identify the key components (operative logics, main foci, and key priorities) of the two disciplines that were inadvertently entering into the problem space of digital well-being. This in turn allowed us to recognize the perspectives on digital well-being produced by our disciplines as partial and clearly driven by the specific orientation of each discipline. Recognizing this partiality meant that we no longer saw the emerging perspectives as defining the identity of the object. Instead, the resulting insights served as material for constructing new dimensions and partial models of digital well-being, as well as identifying crucial blindspots and key questions for further research. Table 3 summarizes our engagement with this method and the resulting hypotheses.

5.1.2 Digital well-being and the practice of fiction. Another method proposed by Schmid to enable further underdetermination of integrative objects is the practice of fiction [95, 97]. As introduced above, this design practice is oriented towards the future, where the

Table 3: Example construction of dimensions of digital well-being through the parameters of disciplines (computer science and critical theory).

	Computer science	Critical theory
Components and parameters of the discipline	Operative logic of quantification, metrification, and optimization. Focus on the computable. Prioritization of measurable outcomes. Deriving value from specificity and efficiency.	Operative logic of critique. Focus on power structures. Prioritization of social, cultural, and historical specificity. Deriving value from demonstrating relevance to concrete problems.
Partial perspectives on digital well-being	Normative (revolving around the algorithmization of 'good life' or 'human good'). Individualized (related to user experience; driven by user interactions with particular devices, platforms, etc.). Quantifiable (can be determined by data collection and engagement metrics).	Discursive (mobilizes different assumptions, worldviews, and agendas; particularly entangled with the discourse of neoliberal capitalism). Culturally situated (dependent on divergent culturally-embedded values). Political (entangled with power dynamics).
Emerging dimensions, partial models, and zones of unknowing	Biopolitical dimension (constructed by metrification as a vector of power). Digital well-being as a scalar object (requires a dynamic model with variables corresponding to different scales - the scale of the individual, of a community, planetary scale, etc.). Which aspects of digital well-being are unmeasurable? Undefinable? Independent from discourse?	

Table 4: Example construction of digital well-being through the practice of fiction.

Procedure	Possible examples of hypothetical perspectives
Removal of one of the components of one's disciplinary training	Critical theory: digital well-being beyond the guiding concept of capitalism. Computer science: digital well-being beyond its computational tractability.
Removal of personal theoretical thresholds that may have structured particular modes of inquiry in the past	A humanities scholar trained in Foucauldian theory: imagining digital well- being beyond a capillary notion of power, or biopolitical subjectivation. A computer science scholar trained in user modeling: imagining digital well- being beyond the lens of user engagement.
Removal of specific properties of the object	Digital well-being without a notion of 'the good'. Digital well-being without the user.

future is not treated as a simple continuation of the present, and objects of design are not approached as fully realized. The future as a generic operator allows for concepts such as digital well-being to be conceived of independently of their current conditions, as partially unknown scenarios projected from the future to the present. The notion of fiction functions here not as a story or a narrative, but as a link between different disciplinary fragments and different temporalities of design, enabling new relations between objects and sets of knowledge. The technique of fiction consists of selecting certain properties of a given discipline or object and reimagining those disciplines or objects without those properties. Examples of this type of thinking can be found in numerous contexts, such as philosophy of science and philosophy of mathematics (science without numbers [28], structuralism without structures [46], mathematics without numbers [45]), or in design theory (a house without a roof [42], a chair without legs [75]). The property we subtract cannot be one of the most essential, otherwise the endeavor will lead to an impasse, nor too insignificant, otherwise the practice will not lead to the creation of new objects. Instead, the initial object, such as digital well-being, is rendered as an unknown object 'X' to

which properties can be added or subtracted, bringing fragmented knowledge together in new formations. The aim is not to try and construct that new object (which would be actively enacting a lack), but to use the hypothetical to mobilize thought in a previously unexplored way. Hypothetically depriving disciplines and objects of one of their significant properties forces us to reorganize knowledge, opening the door for new insights and for designing these objects differently in the future.

Using this approach in relation to our case study, we propose three procedures as a springboard for further expansion of collaborative research on digital well-being. We summarize the procedures and their applications in Table 4.

5.1.3 Future use of these methods. While our own use of these methods have so far been related to critical theory and computer science, the same tools could be applied to other disciplines invested in digital well-being, as well as other integrative objects. For example, the examples proposed in Table 3 can help other researchers engage in the process of disciplinary decomposition and sketch out emerging dimensions of the integrative objects they study. In particular, the first section of the table can help identify

the key parameters of disciplines when entering into a given problem space. By drawing on these prompts, future researchers may wish to ask questions such as: What are the main priorities and operative logics of my discipline? This in turn can help crystallize the partial perspectives produced by disciplines in relation to the studied integrative object: How do the priorities of my discipline influence its view of the object? What perspectives do they produce? Researchers can then gradually outline the emerging dimensions of the object and identify potential new models, hypotheses, and 'zones of unknowing': What remains unknown or unknowable about the object? Similarly, and more attuned to the question of design, the procedures proposed in Table 4 can be also applied in other contexts to offer new perspectives on particular objects of inquiry: What would my perspective on the object be if I subtracted from it an aspect of my disciplinary training? What would my research object be without one of its properties that I take for granted? Our hope is that other researchers might find these questions and methods useful in generating new hypotheses and productive questions for integrative objects in HCI and adjacent fields.

6 CONCLUSION

As a multifaceted discipline with a huge variety of epistemic, analytical, and practical interests, HCI is inevitably concerned with phenomena that defy attempts to impose discrete disciplinary boundaries upon them. Accordingly, this paper has answered a pressing need to better understand how to conduct collaborative, interdisciplinary research that can account for the complexity of contemporary objects of sociotechnical concern. This paper is positioned as a contribution to the existing research on digital well-being, as well as to the ongoing discourse surrounding the challenges of collaborating across disciplines. In line with generic epistemology's key principle of non-dogmatism, and its prioritization of invention over negation, we do not aim to critique existing scholarship, rather we have proposed an alternative perspective, rooted in our own experiences, and guided by the framework developed by Schmid and her collaborators. The aim of this approach, put simply, is for researchers to recognize their disciplinary blindspots and leave their disciplinary identity at the door as they enter a collaboration. This process seeks to strip away any implicit assumptions about the object of inquiry brought into the research space. To this end, this paper has provided practical recommendations for researchers and practitioners in the HCI community and beyond to build on in the future, in the hope of facilitating non-hierarchical and non-dogmatic modes of knowledge expansion.

The recommendations we have proposed in this paper are by no means definitive or exhaustive; they are meant more as an invitation to further experimentation and discussion. Our intention is to develop a research space where new paradigms of thinking can flourish and where integrative objects, such as digital well-being, can be re-imagined and re-designed according to collectively established priorities, rather than unilaterally imposed protocols. While ideas of disciplinary forgetting, dissolution, and subtraction might be possible to grapple with conceptually, putting generic epistemology to practice can be seen as a balancing act. The need to publish the results of a collaboration as a paper within a discipline, or to explain findings to disciplinary colleagues, often pull

researchers out of the space of the generic. We might need to become comfortable producing outcomes unintelligible to our own home disciplines. As such, we see this current paper as taking a step towards a more extensive exploration of generic epistemology as a generative approach for engaging with sociotechnical objects in HCI and other related fields.

REFERENCES

- 2008. Statistics vs. Machine Learning, fight! Retrieved 2022-12-09 from https://brenocon.com/blog/2008/12/statistics-vs-machine-learning-fight/
- [2] 2022. CHI '22: Proceedings of the 2022 CHI Conference on Human Factors in Computing Systems (New Orleans, LA, USA). Association for Computing Machinery, New York, NY, USA.
- [3] Solon Barocas, Moritz Hardt, and Arvind Narayanan. 2019. Fairness and Machine Learning: Limitations and Opportunities. fairmlbook.org. http://www.fairmlbook.org.
- [4] Andrew Barry, Georgina Born, and Gisa Weszkalnys. 2008. Logics of interdisciplinarity. Economy and Society 37, 1 (Feb. 2008), 20–49. https://doi.org/10.1080/ 03085140701760841
- [5] Eric P. S. Baumer and Jed R. Brubaker. 2017. Post-Userism. In Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems (CHI '17). Association for Computing Machinery, New York, NY, USA, 6291–6303. https://doi.org/10.1145/3025453.3025740 event-place: Denver, Colorado, USA.
- [6] Eleonora Belfiore. 2015. 'Impact', 'value' and 'bad economics': Making sense of the problem of value in the arts and humanities. Arts and Humanities in Higher Education 14, 1 (Feb. 2015), 95–110. https://doi.org/10.1177/1474022214531503
- [7] Mario Biagioli. 2009. Postdisciplinary Liaisons: Science Studies and the Humanities. Critical Inquiry 35, 4 (Jan. 2009), 816–833. https://doi.org/10.1086/599586
- [8] Alan F. Blackwell. 2015. Filling the Big Hole in HCI Research. Interactions 22, 6 (oct 2015), 37–41. https://doi.org/10.1145/2830317
- [9] Alan F. Blackwell. 2015. HCI as an Inter-Discipline. In Proceedings of the 33rd Annual ACM Conference Extended Abstracts on Human Factors in Computing Systems (CHI EA '15). Association for Computing Machinery, New York, NY, USA, 503–516. https://doi.org/10.1145/2702613.2732505
- [10] Geoffrey C. Bowker and Susan Leigh Star. 2008. Sorting things out: classification and its consequences (1. paperback ed., 8. print ed.). MIT Press, Cambridge, Mass.
- [11] Richard Buchanan. 1992. Wicked Problems in Design Thinking. Design Issues 8, 2 (1992), 5–21. https://doi.org/10.2307/1511637 Publisher: The MIT Press.
- [12] Christopher Burr and Luciano Floridi (Eds.). 2020. Ethics of digital well-being: a multidisciplinary approach. Springer, Cham.
- [13] Christopher Burr, Mariarosaria Taddeo, and Luciano Floridi. 2020. The Ethics of Digital Well-Being: A Thematic Review. Science and Engineering Ethics 26, 4 (Aug. 2020), 2313–2343. https://doi.org/10.1007/s11948-020-00175-8
- [14] Danilo Bzdok, Naomi Altman, and Martin Krzywinski. 2018. Statistics versus Machine Learning. Nature Methods 15 (2018), 233–234.
- [15] Carrie J Cai, Samantha Winter, David Steiner, Lauren Wilcox, and Michael Terry. 2021. Onboarding Materials as Cross-Functional Boundary Objects for Developing AI Assistants. In Extended Abstracts of the 2021 CHI Conference on Human Factors in Computing Systems (Yokohama, Japan) (CHI EA '21). Association for Computing Machinery, New York, NY, USA, Article 43, 7 pages. https://doi.org/10.1145/3411763.3443435
- [16] John M. Carroll. 2010. Conceptualizing a possible discipline of human-computer interaction. *Interacting with Computers* 22, 1 (2010), 3–12. https://doi.org/10. 1016/j.intcom.2009.11.008 Special Issue: Festschrift for John Long.
- [17] Marta E. Cecchinato, John Rooksby, Alexis Hiniker, Sean Munson, Kai Lukoff, Luigina Ciolfi, Anja Thieme, and Daniel Harrison. 2019. Designing for Digital Wellbeing: A Research & Practice Agenda. In Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems. ACM, Glasgow Scotland Uk, 1–8. https://doi.org/10.1145/3290607.3298998
- [18] Léo Coutellec. 2019. Penser Alzheimer comme un objet (dés)intégratif. Contribution épistémologique et enjeux éthiques. In L'aventure épistémologique contemporaine: rencontres autour des travaux d'Anne-Françoise Schmid, Paris Chrysos and Annie Gentès (Eds.). Éditions Kimé, Paris IIe, 137–152.
- [19] Anna Croon. 2022. Thinking with care in human-computer interaction. Feminist Theory 23, 2 (April 2022), 232–246. https://doi.org/10.1177/14647001221082294 Publisher: SAGE Publications.
- [20] Nigel Cross. 1982. Designerly ways of knowing. Design Studies 3, 4 (Oct. 1982), 221–227. https://doi.org/10.1016/0142-694X(82)90040-0
- [21] Matthew J. Dennis. 2021. Digital well-being under pandemic conditions: catalysing a theory of online flourishing. Ethics and Information Technology 23, 3 (Sept. 2021), 435–445. https://doi.org/10.1007/s10676-021-09584-0
- [22] Alan Dix. 2010. Human-computer interaction: A stable discipline, a nascent science, and the growth of the long tail. *Interacting with Computers* 22, 1 (2010), 13–27. https://doi.org/10.1016/j.intcom.2009.11.007 Special Issue: Festschrift

- for John Long.
- [23] Niall Docherty. 2020. Facebook's Ideal User: Healthy Habits, Social Capital, and the Politics of Well-Being Online. Social Media + Society 6, 2 (2020). https://doi.org/10.1177/2056305120915606
- [24] Niall Docherty. 2021. Digital Self-Control and the Neoliberalization of Social Media Well-Being. *International Journal of Communication* 15 (Aug. 2021), 3827–3846. https://ijoc.org/index.php/ijoc/article/view/17721/3532
- [25] Paul Dourish, Janet Finlay, Phoebe Sengers, and Peter Wright. 2004. Reflective HCI: towards a critical technical practice. In Extended abstracts of the 2004 conference on Human factors and computing systems - CHI '04. ACM Press, Vienna, Austria, 1727. https://doi.org/10.1145/985921.986203
- [26] Daniel Fallman. 2003. Design-oriented human-computer interaction. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '03). Association for Computing Machinery, New York, NY, USA, 225–232. https://doi.org/10.1145/642611.642652
- [27] Paul Feyerabend. 1975. Against method: outline of an anarchistic theory of knowledge. NLB; Humanities Press, London: Atlantic Highlands.
- [28] Hartry Field. 2016. Science without numbers: a defense of nominalism (second ed.). Oxford University Press, Oxford, United Kingdom.
- [29] Robert Frodeman and Carl Mitcham. 2007. New Directions in Interdisciplinarity: Broad, Deep, and Critical. Bulletin of Science, Technology & Society 27, 6 (2007), 506–514. https://doi.org/10.1177/0270467607308284
- [30] Peter Galison. 1997. Image and logic: a material culture of microphysics. University of Chicago Press, Chicago.
- [31] Annie Gentès. 2019. Inventions technique et "objet intégratif". In L'aventure épistémologique contemporaine: rencontres autour des travaux d'Anne-Françoise Schmid, Paris Chrysos and Annie Gentès (Eds.). Éditions Kimé, Paris IIe, 153– 176.
- [32] Michael Gibbons, Camille Limoges, Helga Nowotny, Simon Schwartzman, Peter Scott, and Martin Trow. 1994. The New Production of Knowledge: The Dynamics of Science and Research in Contemporary Societies. SAGE Publications Ltd, London. http://digital.casalini.it/9781446265871
- [33] Google. 2022. Expanding research on digital wellbeing. https://blog.google/technology/health/expanding-research-on-digital-wellbeing/
- [34] Michael E. Gorman. 2004. Collaborating on Convergent Technologies: Education and Practice. Annals of the New York Academy of Sciences 1013, 1 (May 2004), 25–37. https://doi.org/10.1196/annals.1305.003
- [35] Michael E. Gorman and Jim Spohrer. 2010. 75Service Science: A New Expertise for Managing Sociotechnical Systems. In Trading Zones and Interactional Expertise: Creating New Kinds of Collaboration. The MIT Press. https://doi.org/ 10.7551/mitpress/9780262014724.003.0005 arXiv:https://academic.oup.com/mitpress-scholarship-online/book/0/chapter/168376182/chapter-agpdf/44891786/book_14407_section_168376182.ag.pdf
- [36] UK Government. 2020. Online Harms White Paper: Full government response to the consultation. https://www.gov.uk/government/consultations/onlineharms-white-paper/outcome/online-harms-white-paper-full-governmentresponse. https://www.gov.uk/government/consultations/online-harms-whitepaper/outcome/online-harms-white-paper-full-government-response
- [37] Nina Grgic-Hlaca, Elissa M. Redmiles, Krishna P. Gummadi, and Adrian Weller. 2018. Human Perceptions of Fairness in Algorithmic Decision Making: A Case Study of Criminal Risk Prediction. In Proceedings of the 2018 World Wide Web Conference (Lyon, France) (WWW '18). International World Wide Web Conferences Steering Committee, Republic and Canton of Geneva, CHE, 903–912. https://doi.org/10.1145/3178876.3186138
- [38] Jonathan Grudin. 1990. The Computer Reaches out: The Historical Continuity of Interface Design. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '90). Association for Computing Machinery, New York, NY, USA, 261–268. https://doi.org/10.1145/97243.97284 event-place: Seattle, Washington, USA.
- [39] Marco Gui, Marco Fasoli, and Roberto Carradore. 2017. 'Digital Well-Being'. Developing a New Theoretical Tool For Media Literacy Research. *Italian Journal of Sociology of Education* 9, 02/2017 (2017), 155–173. https://doi.org/10.14658/puni-iise-2017-1-8
- [40] Donna Jeanne Haraway. 2008. When species meet. Number 3 in Posthumanities. University of Minnesota Press, Minneapolis. OCLC: ocn156975211.
- [41] Richard H. R. Harper. 2019. The Role of HCI in the Age of AI. International Journal of Human-Computer Interaction 35, 15 (Sept. 2019), 1331–1344. https://doi.org/10.1080/10447318.2019.1631527 Publisher: Taylor & Francis _eprint: https://doi.org/10.1080/10447318.2019.1631527.
- [42] Armand Hatchuel, Pascal Le Masson, and Benoit Weil. 2017. C-K Theory: Modelling Creative Thinking and Its Impact on Research. In Creativity, Design Thinking and Interdisciplinarity, Frédéric Darbellay, Zoe Moody, and Todd Lubart (Eds.). Springer Singapore, Singapore, 169–183. https://doi.org/10.1007/978-981-10-7524-7 11
- [43] Armand Hatchuel and Benoît Weil. 2008. Les nouveaux régimes de la conception: langages, théories, métiers [actes du colloque de Cerisy]. Vuibert Cerisy, Paris.

- [44] Armand Hatchuel and Benoit Weil. 2009. C-K design theory: an advanced formulation. Research in Engineering Design 19, 4 (Jan. 2009), 181–192. https://doi.org/10.1007/s00163-008-0043-4
- [45] Geoffrey Hellman. 1989. Mathematics without numbers: towards a modalstructural interpretation. Clarendon, Oxford.
- [46] Geoffrey Hellman. 1996. Structuralism Without Structures. Philosophia Mathematica 4, 2 (May 1996), 100–123. https://doi.org/10.1093/philmat/4.2.100
- [47] Flis Henwood and Benjamin Marent. 2019. Understanding digital health: Productive tensions at the intersection of sociology of health and science and technology studies. Sociology of Health & Illness 41, S1 (Oct. 2019), 1–15. https://doi.org/10.1111/1467-9566.12898
- [48] Kenneth Holstein, Jennifer Wortman Vaughan, Hal Daumé, Miro Dudik, and Hanna Wallach. 2019. Improving Fairness in Machine Learning Systems: What Do Industry Practitioners Need?. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems (Glasgow, Scotland Uk) (CHI '19). Association for Computing Machinery, New York, NY, USA, 1–16. https://doi. org/10.1145/3290605.3300830
- [49] Matthew K. Hong, Adam Fourney, Derek DeBellis, and Saleema Amershi. 2021. Planning for Natural Language Failures with the AI Playbook. In Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems (Yokohama, Japan) (CHI '21). Association for Computing Machinery, New York, NY, USA, Article 386, 11 pages. https://doi.org/10.1145/3411764.3445735
- [50] Clare J. Hooper, David Millard, Jill Fantauzzacoffin, and Joseph 'Jofish' Kaye. 2013. Science vs. Science: The Complexities of Interdisciplinary Research. In CHI '13 Extended Abstracts on Human Factors in Computing Systems (CHI EA '13). Association for Computing Machinery, New York, NY, USA, 2541–2544. https://doi.org/10.1145/2468356.2468825
- [51] Lucien Hounkpatin, Henny Wexler-Czitrom, Avner Perez, and Lauriane Courbin. 2011. Chapitre 3. Vers un nouveau paradigme: la clinique de la multiplicité et la fabrique de « l'intime collectif ». In Comprendre et traiter les situations interculturelles. De Boeck Supérieur, 55. https://doi.org/10.3917/dbu.pirlo.2011. 01.0055
- [52] Katri Huutoniemi, Julie Thompson Klein, Henrik Bruun, and Janne Hukkinen. 2010. Analyzing interdisciplinarity: Typology and indicators. Research Policy 39, 1 (Feb. 2010), 79–88. https://doi.org/10.1016/j.respol.2009.09.011
- [53] Wall Street Journal. 2021. The Facebook Files. https://www.wsj.com/articles/the-facebook-files-11631713039. https://www.wsj.com/articles/the-facebook-files-11631713039
- [54] Karen Kastenhofer, Ulrike Bechtold, and Harald Wilfing. 2011. Sustaining sustainability science: The role of established inter-disciplines. *Ecological Economics* 70, 4 (Feb. 2011), 835–843. https://doi.org/10.1016/j.ecolecon.2010.12.008
- [55] Julie Thompson Klein. 2017. Typologies of Interdisciplinarity: The Boundary Work of Definition. In The Oxford Handbook of Interdisciplinarity (second ed.), Robert Frodeman, Julie Thompson Klein, and Roberto C. S. Pacheco (Eds.). Oxford University Press, Oxford, United Kingdom, 21–34.
- [56] Thomas S. Kuhn. 2012. The Structure of Scientific Revolutions (fourth ed.). The University of Chicago Press, Chicago; London.
- [57] Stacey Kuznetsov and Martin Tomitsch. 2018. A Study of Urban Heat: Understanding the Challenges and Opportunities for Addressing Wicked Problems in HCI. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems (CHI '18). Association for Computing Machinery, New York, NY, USA, 1–13. https://doi.org/10.1145/3173574.3174137
- [58] Cherie Lacey, Alex Beattie, and Catherine Caudwell. 2021. Wellness Capitalism and the Design of the Perfect User. *Interface critique* 3 (Sept. 2021), 127–150. https://doi.org/10.11588/IC.2021.3.81323
- [59] François Laruelle. 2008. Introduction aux sciences génériques. Éditions Pétra, Paris.
- [60] François Laruelle. 2010. Philosophie non-standard: générique, quantique, philofiction. Kimé, Paris.
- [61] François Laruelle. 2013. Philosophy and non-philosophy. Univocal, Minneapolis. OCLC: ocn816563912.
- $\begin{tabular}{ll} [62] François Laruelle. 2013. {\it Principles of non-philosophy}. Bloomsbury, New York. \\ \end{tabular}$
- [63] Bruno Latour. 2007. Reassembling the social: an introduction to Actor-Network-Theory (first ed.). Oxford Univ. Press, Oxford.
- [64] Min Kyung Lee, Daniel Kusbit, Anson Kahng, Ji Tae Kim, Xinran Yuan, Allissa Chan, Daniel See, Ritesh Noothigattu, Siheon Lee, Alexandros Psomas, and Ariel D. Procaccia. 2019. WeBuildAI: Participatory Framework for Algorithmic Governance. Proc. ACM Hum.-Comput. Interact. 3, CSCW, Article 181 (nov 2019), 35 pages. https://doi.org/10.1145/3359283
- [65] Sung-Chul Lee, Jihee Kim, and Juho Kim. 2018. Micro-NGO: Tackling Wicked Social Problems with Problem Solving and Action Planning Support in Chat. In Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems (CHI EA '18). Association for Computing Machinery, New York, NY, USA, 1-6. https://doi.org/10.1145/3170427.3188564
- [66] Jean-Marie Legay. 2006. L'interdisciplinarité dans les sciences de la vie. Éditions Quæ, Versailles. https://www.cairn.info/l-interdisciplinite-dans-les-sciencesde-la-vie--9782738012104.htm

- [67] Jean-Marie Legay and Anne-Françoise Schmid. 2004. Philosophie de l'interdisciplinarité. Correspondance (1999-2004) sur la recherche scientifique, la modélisation et les objets complexes. Éditions Pétra. https://www.editionspetra.fr/livres/philosophie-de-linterdisciplinaritecorrespondance-1999-2004-sur-la-recherche-scientifique-la
- [68] Susan Leigh Star. 2010. This is Not a Boundary Object: Reflections on the Origin of a Concept. Science, Technology, & Human Values 35, 5 (Sept. 2010), 601-617. https://doi.org/10.1177/0162243910377624
- [69] Tilmann Lindberg, Christine Noweski, and Christoph Meinel. 2010. Evolving discourses on design thinking: how design cognition inspires meta-disciplinary creative collaboration. Technoetic Arts 8, 1 (May 2010), 31-37. https://doi.org/ 10.1386/tear.8.1.31/1
- [70] John Long and John Dowell. 1989. Conceptions of the discipline of HCI: Craft, applied science, and engineering. In HCI, Vol. 89. 9-32.
- Robin Mackay. 2012. On Making Ready. In Reprototypes, Triangulations and Road Tests, Eva Wilson and Daniela Zyman (Eds.). Sternberg Press, 15-21.
- [72] Robin Mackay. 2019. Exposition/Exercice: Propositions sur l'Objet Intégratif de l'Art Contemporain. In L'aventure épistémologique contemporaine: rencontres autour des travaux d'Anne-Françoise Schmid, Paris Chrysos and Annie Gentès (Eds.). Éditions Kimé, Paris IIe, 177-191.
- [73] Miles MacLeod. 2018. What makes interdisciplinarity difficult? Some consequences of domain specificity in interdisciplinary practice. Synthese 195, 2 (Feb. 2018), 697-720. https://doi.org/10.1007/s11229-016-1236-4
- Muriel Mambrini-Doudet. 2019. Une épistémologie dans l'intimité du futur des sciences. In L'aventure épistémologique contemporaine: rencontres autour des travaux d'Anne-Françoise Schmid, Paris Chrysos and Annie Gentès (Eds.). Éditions Kimé, Paris IIe, 207-233.
- Pascal Le Masson and Chris Mcmahon, 2016. Armand Hatchuel et Benoit Weil La théorie C-K, un fondement formel aux théories de l'innovation. In Les grands auteurs du management de l'innovation et de la créativité, Thierry Burger-Helmchen, Caroline Hussler, and Patrick Cohendet (Eds.). In Quarto -Editions Management et Société, 588-613. https://hal-mines-paristech.archivesouvertes.fr/hal-01243331
- Gale Moore and Danielle Lottridge. 2010. Interaction design in the university: designing disciplinary interactions. In CHI '10 Extended Abstracts on Human Factors in Computing Systems. ACM, Atlanta Georgia USA, 2735-2744. https: //doi.org/10.1145/1753846.1753858
- [77] Deirdre K. Mulligan, Joshua A. Kroll, Nitin Kohli, and Richmond Y. Wong. 2019. This Thing Called Fairness: Disciplinary Confusion Realizing a Value in Technology. Proc. ACM Hum.-Comput. Interact. 3, CSCW, Article 119 (nov 2019), 36 pages. https://doi.org/10.1145/3359221
- [78] Helga Nowotny, Peter Scott, and Michael Gibbons. 2011. Re-Thinking science: knowledge and the public in an age of uncertainty (repr ed.). Polity Press, Cam-
- Daniel Oberhaus. 2019. Extraterrestrial languages. The MIT Press, Cambridge, Massachusetts; London England.
- [80] Michael O'Rourke and Stephen J. Crowley. 2013. Philosophical intervention and cross-disciplinary science: the story of the Toolbox Project. Synthese 190, 11 (July 2013), 1937-1954. https://doi.org/10.1007/s11229-012-0175-y
- [81] David Budtz Pedersen. 2016. Integrating social sciences and humanities in interdisciplinary research. Palgrave Communications 2, 1 (July 2016), 1-7. https: //doi.org/10.1057/palcomms.2016.36
- Sachin R Pendse, Daniel Nkemelu, Nicola J Bidwell, Sushrut Jadhav, Soumitra Pathare, Munmun De Choudhury, and Neha Kumar. 2022. From Treatment to Healing:Envisioning a Decolonial Digital Mental Health. In Proceedings of the 2022 CHI Conference on Human Factors in Computing Systems (CHI '22). Association for Computing Machinery, New York, NY, USA. https://doi.org/10. 1145/3491102.3501982
- [83] James Pierce, Phoebe Sengers, Tad Hirsch, Tom Jenkins, William Gaver, and Carl DiSalvo. 2015. Expanding and Refining Design and Criticality in HCI. In Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems (CHI '15). Association for Computing Machinery, New York, NY, USA, 2083-2092. https://doi.org/10.1145/2702123.2702438
- [84] Mary Louise Pratt. 1991. Arts of the Contact Zone. Profession (1991), 33-40. https://www.jstor.org/stable/25595469 Publisher: Modern Language Associa-
- Stuart Reeves. 2015. Human-Computer Interaction as Science. In Proceedings of The Fifth Decennial Aarhus Conference on Critical Alternatives (Aarhus, Denmark) (CA '15). Aarhus University Press, Aarhus N, 73-84. https://doi.org/10.7146/ aahcc.v1i1.21296
- Stuart Reeves. 2015. Locating the 'big Hole' in HCI Research. Interactions 22, 4 (jun 2015), 53-56. https://doi.org/10.1145/2785986
- BBC Research and Development. 2022. Digital Wellbeing. https://www.bbc.co. uk/rd/projects/digital-wellbeing Horst W. J. Rittel and Melvin M. Webber. 1973. Dilemmas in a general theory of
- planning. Policy Sciences 4 (1973), 155-169.

- [89] Elizabeth B.-N. Sanders and Bo Westerlund. 2011. Experiencing, Exploring and Experimenting in and with Co-Design Spaces. Nordes 4 (March 2011). https://archive.nordes.org/index.php/n13/article/view/110 Number: 4.
- [90] Martin Savransky. 2021. Problems All the Way Down. Theory, Culture & Society 38, 2 (March 2021), 3-23. https://doi.org/10.1177/0263276420966389 Publisher: SAGE Publications Ltd.
- [91] Anne-Françoise Schmid. 1998. L'âge de l'épistémologie: science, ingénierie, éthique. Kimé, Paris.
- Anne-Françoise Schmid. 2001. Henri Poincaré: les sciences et la philosophie (2e éd. ed.). L'Harmattan, Paris
- Anne-Françoise Schmid. 2001. Que peut la philosophie des sciences. Pétra, Paris.
- Anne-Françoise Schmid. 2015. On Contemporary Objects. In Simulation, Exercise, Operations, Robin Mackay (Ed.). Urbanomic, 63-68
- Anne-Françoise Schmid. 2018. The Philosophical Underpinnings of Design Theory. In Advancements in the Philosophy of Design, Pieter E. Vermaas and Stéphane Vial (Eds.). Springer International Publishing, Cham, 415-430. https:// //doi.org/10.1007/978-3-319-73302-9_19
- [96] Anne-Françoise Schmid. 2021. Scripts philosophiques 1: Le Silence du Futur. Chisokudo Publications, Nagoya, Japan.
- Anne-Françoise Schmid and Armand Hatchuel. 2014. On Generic Epistemology. Angelaki 19, 2 (April 2014), 131-144. https://doi.org/10.1080/0969725X.2014.
- [98] Anne-Françoise Schmid and Muriel Mambrini-Doudet. 2019. Épistémologie générique: manuel pour les sciences futures. Éditions Kimé.
- Anne-Françoise Schmid, Muriel Mambrini-Doudet, and Armand Hatchuel. 2011. Une nouvelle logique de l'interdisciplinarité. Nouvelles perspectives en sciences sociales 7, 1 (Dec. 2011), 105-136. https://doi.org/10.7202/1007084ar
- [100] Frank M. Schneider, Sarah Lutz, Annabell Halfmann, Adrian Meier, and Leonard Reinecke, 2022. How and when do mobile media demands impact well-being? Explicating the Integrative Model of Mobile Media Use and Need Experiences (IM ³ UNE). Mobile Media & Communication 10, 2 (May 2022), 251–271. https: //doi.org/10.1177/20501579211054928
- [101] Ben Schouten, Gwen Klerks, Marcel Den Hollander, and Nicolai Brodersen Hansen. 2021. Action Design Research Shaping University-Industry Collaborations for Wicked Problems. In Proceedings of the 32nd Australian Conference on Human-Computer Interaction (OzCHI '20). Association for Computing Machinery, New York, NY, USA, 36-44. https://doi.org/10.1145/3441000.3441078
- [102] Oscar Schwartz. 2019. Why beating your phone addiction may come at a cost. The Guardian (March 2019). https://www.theguardian.com/technology/2019/ mar/13/digital-wellness-phone-addiction-tech
- [103] Andrew D. Selbst, Danah Boyd, Sorelle A. Friedler, Suresh Venkatasubramanian, and Janet Vertesi. 2019. Fairness and Abstraction in Sociotechnical Systems. In Proceedings of the Conference on Fairness, Accountability, and Transparency (Atlanta, GA, USA) (FAT* '19). Association for Computing Machinery, New York, NY, USA, 59-68. https://doi.org/10.1145/3287560.3287598
- [104] Andrew D Selbst, Danah Boyd, Sorelle A Friedler, Suresh Venkatasubramanian, and Janet Vertesi. 2019. Fairness and abstraction in sociotechnical systems. In Proceedings of the conference on fairness, accountability, and transparency. 59-68.
- [105] Susan Leigh Star and James R. Griesemer. 1989. Institutional Ecology, 'Translations' and Boundary Objects: Amateurs and Professionals in Berkeley's Museum of Vertebrate Zoology, 1907-39. Social Studies of Science 19, 3 (1989), 387-420. http://www.jstor.org/stable/285080
- [106] Christopher Starke, Janine Baleis, Birte Keller, and Frank Marcinkowski. 2022. Fairness perceptions of algorithmic decision-making: A systematic review of the empirical literature. Big Data & Society 9, 2 (2022), 20539517221115189. https://doi.org/10.1177/20539517221115189 arXiv:https://doi.org/10.1177/20539517221115189
- [107] Chad J. Valasek. 2022. Disciplining the Akratic user: Constructing digital (un) //doi.org/10.1177/20501579211038796
- Mariek M P Vanden Abeele. 2021. Digital Wellbeing as a Dynamic Construct. Communication Theory 31, 4 (Nov. 2021), 932-955. https://doi.org/10.1093/ct/
- [109] Mariek M. P. Vanden Abeele and Minh Hao Nguyen. 2022. Digital well-being in an age of mobile connectivity: An introduction to the Special Issue. Mobile Media & Communication 10, 2 (May 2022), 174-189. https://doi.org/10.1177/ 20501579221080899
- [110] Sandra Wachter, Brent Mittelstadt, and Chris Russell. 2021. Why fairness cannot be automated: Bridging the gap between EU non-discrimination law and AI. Computer Law & Security Review 41 (2021), 105567. https://doi.org/10.1016/j. clsr.2021.105567
- [111] Allison Woodruff, Sarah E. Fox, Steven Rousso-Schindler, and Jeffrey Warshaw. 2018. A Qualitative Exploration of Perceptions of Algorithmic Fairness. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems (Montreal OC, Canada) (CHI '18). Association for Computing Machinery, New York, NY, USA, 1-14. https://doi.org/10.1145/3173574.3174230

- [112] Peter Wright, Mark Blythe, and John McCarthy. 2006. User Experience and the Idea of Design in HCI. In Interactive Systems. Design, Specification, and Verification (Lecture Notes in Computer Science), Stephen W. Gilroy and Michael D. Harrison (Eds.). Springer, Berlin, Heidelberg, 1–14. https://doi.org/10.1007/ 11752707_1
- [113] Daisy Yoo, Odeth Kantengwa, Nick Logler, Reverien Interayamahanga, Joseph Nkurunziza, and Batya Friedman. 2018. Collaborative Reflection: A Practice for Enriching Research Partnerships Spanning Culture, Discipline, and Time. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems. ACM, Montreal QC Canada, 1–11. https://doi.org/10.1145/3173574.3173853
- [114] Kimberly Young. 2009. Internet Addiction: Diagnosis and Treatment Considerations. Journal of Contemporary Psychotherapy 39, 4 (Dec. 2009), 241–246. https://doi.org/10.1007/s10879-009-9120-x
- [115] Clint Zeagler, Maribeth Gandy, Scott Gilliland, Delton Moore, Rocco Centrella, and Brandon Montgomery. 2017. In Harmony: Making a Wearable Musical
- Instrument as a Case Study of Using Boundary Objects in an Interdisciplinary Collaborative Design Process. In *Proceedings of the 2017 Conference on Designing Interactive Systems* (Edinburgh, United Kingdom) (*DIS '17*). Association for Computing Machinery, New York, NY, USA, 369–378. https://doi.org/10.1145/3064663.3064678
- [116] Xiaomu Zhou, Mark Ackerman, and Kai Zheng. 2011. CPOE Workarounds, Boundary Objects, and Assemblages. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (Vancouver, BC, Canada) (CHI '11). Association for Computing Machinery, New York, NY, USA, 3353–3362. https: //doi.org/10.1145/1978942.1979439
- [117] John Zimmerman, Jodi Forlizzi, and Shelley Evenson. 2007. Research through Design as a Method for Interaction Design Research in HCI. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '07). Association for Computing Machinery, New York, NY, USA, 493–502. https: //doi.org/10.1145/1240624.1240704 event-place: San Jose, California, USA.